HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.

HO M E W O R K 1 E C S 2 3 5 � S P R I N G 2 0 0 3

Version of April 1, 2003 11:45 am Page 1 of 1

Homework 1

Due Date: April 7, 2003 Points: 10

This assignment tries to put you in the right frame of mind for the course, which will require you to �nd out what assumptions are being made, and then question them. Please select any work of �ction (such as a book, movie, or poem) that either explicitly states some assumption, or clearly expects or causes the reader (or viewer) to make some assumption, that is invalid by the end of the work. The work may discuss exploiting a weakness in a system, or just take an unusual view of a situation that reveals something previously unknown, or that shows the situation is not as it seems. Please write at most 1 page about the twist and how it relates to the assumptions that characters in the work (or targets of the twist) make, as well as how the twist illustrates the (mis)placement of trust in the assumptions.